Title: Menetrier's Disease Presenting as Recurrent Unprovoked Venous Thrombosis

Authors:
Karl Greenblatt (hkvmg15@gmail.com)
Brave Nguyen (nguyenbr@ucmail.uc.edu)

Version: 2 Date: 01 Dec 2018

Author’s response to reviews:

The discussion section has been expanded further, specifically with regard to:

- A review of normal secondary hemostasis, prior to discussing pathophysiology
- The individual lab abnormalities in our patient and what contributed to each
- The significance of the positive antiphospholipid antibody
- The significance of the patient's history of juvenile polyposis syndrome
- The most unusual and noteworthy aspects of the case